Author’s response to reviews

Title: Few women received a specific explanation of stillbirth - an online survey of women’s perceptions and thoughts about the cause of their baby’s death

Authors:

Berit Höglund (berit.hoglund@kbh.uu.se)

Ingela Rådestad (ingela.radestad@shh.se)

Ingegerd Hildingsson (ingegerd.hildingsson@kbh.uu.se)

Version: 5 Date: 10 Apr 2019

Author’s response to reviews:

The grammatical errors are revised as suggested.